IZA DP No. 296

Intergenerational Transmissions and the Schoolto-Work Transition of 2nd Generation Immigrants

Helena Skyt Nielsen Michael Rosholm Nina Smith Leif Husted

May 2001

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Intergenerational Transmissions and the Schoolto-Work Transition of 2nd Generation Immigrants

Helena Skyt Nielsen

CIM and Aarhus School of Business

Michael Rosholm

University of Aarhus, CIM, and IZA, Bonn

Nina Smith

Aarhus School of Business, CIM, and IZA, Bonn

Leif Husted

Institute of Local Government Studies and CIM

Discussion Paper No. 296 May 2001

IZA

P.O. Box 7240 D-53072 Bonn Germany

Tel.: +49-228-3894-0 Fax: +49-228-3894-210 Email: iza@iza.org

This Discussion Paper is issued within the framework of IZA's research area *Mobility and Flexibility of Labor Markets*. Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent, nonprofit limited liability company (Gesellschaft mit beschränkter Haftung) supported by the Deutsche Post AG. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public. The current research program deals with (1) mobility and flexibility of labor markets, (2) internationalization of labor markets in transition, (5) the future of work, (6) project evaluation and (7) general labor economics.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character.

IZA Discussion Paper No. 296 May 2001

ABSTRACT

Intergenerational Transmissions and the School-to-Work Transition of 2nd Generation Immigrants^{*}

We analyse the extent of intergenerational transmission through parental capital, ethnic capital and neighbourhood effects on several aspects of the school-to-work transition of 2 nd generation immigrants and young ethnic Danes. The main findings are that parental capital has strong positive effects on the probability of completing a qualifying education and on the entry into the labour market, but it has a much smaller impact on the duration of the first employment spell and on the wage level. Growing up in neighbourhoods with a high concentration of immigrants is associated with negative labour market prospects both for young natives and 2 nd generation immigrants.

JEL Classification: J61, J71

Keywords: School-to-work transition, second generation immigrants, intergenerational transmission, parental capital, ethnic capital, neighbourhood effects

Nina Smith Aarhus School of Business Department of Economics and CIM Fuglesangs Alle 20 DK-8210 Aarhus V Tel.: +45 8948 6413 Fax: +45 8615 5175 Email: nina@asb.dk

^{*} Thanks to Thomas Haugaard Jensen, Anne-Sofie Reng Rasmussen and Mette Skak-Nielsen who have done part of the computational work. We also want to thank participants in the ESF-conference on Migration and Development in Naples for many valuable comments. The project has been financed by the Danish Social Research Council (FREJA), Danish Ministry of Labour and TSER.

1. Introduction

In a survey by OECD (1999), Denmark is pointed out as *the* country with the highest relative unemployment rate for (first generation) immigrants among the OECD-countries studied. While immigrants in the US, Canada and Australia experience unemployment rates approximately equalling those of the native born population, immigrants in Denmark have unemployment rates which are more than three times the level of the native population. Although the figures are less depressing when looking at 2nd generation immigrants compared to the native population, 2nd generation immigrants, see Hummelgård et al. (1998). Since the number of adult 2nd generation immigrants grows dramatically during the next decades, it is of primary importance to investigate which factors are associated with a successful labour market entry of different groups of 2nd generation immigrants. Such an investigation is exactly the purpose of this paper.

We analyse the educational attainment and employment for 2^{nd} generation immigrants. We focus on potential intergenerational transmission mechanisms and the extent of intergenerational mobility. Specifically, we try to identify parental and ethnic capital effects separately from neighbourhood effects that may arise for 2^{nd} generation immigrants who grow up in areas or municipalities with a very high ethnic concentration. The models for labour market success of 2^{nd} generation immigrants are compared to parallel models for young ethnic Danes.¹

Four aspects of the school-to-work transition are analysed. First, we study which factors explain whether or not young 2nd generation immigrants obtain a qualifying education by estimating a binary choice model. Second, the waiting time from leaving school to the first ordinary job is analysed in a duration model framework, and the duration of the first employment spell is analysed within that framework as well. Finally, the hourly wage rate in 1997 is analysed by estimating a standard human capital wage function. To account for the intergenerational transmission effects, we include parental capital, ethnic capital and neighbourhood effects in the models. The analyses are based on two panel databases, both arising from administrative registers and covering each of the years 1985-1997. The first sample contains the total population of 1st and 2nd generation immigrants in Denmark, while the second is a random 10% sample of the ethnic Danish population. We restrict the samples to include only individuals aged 35 or less. We are able to identify the fathers and mothers of these young individuals in both data sets and therefore, we have extensive information on family background.

¹ We use the term 'ethnic Danes' for individuals who are not 1st or 2nd generation immigrants.

The results show that intergenerational spillovers exert some influence on the school-to-work transition for the 2nd generation immigrant youth. The intergenerational transmission exerts a strong effect on the probability of obtaining a qualifying education and on entering the first job but after that, the direct influence of intergenerational transmission dies out. There is still a strong indirect effect, since intergenerational spillovers are strongly present in the education decision, and we show that education has a strong influence on employment duration and wages. For females, parental capital is the dominant intergenerational transmission channel, while ethnic capital and neighbourhood effects are important channels as well for males.

If the 2^{nd} generation immigrants succeed in obtaining a qualifying education, the waiting time until they get their first job and the probability of leaving employment are reduced significantly. Thus, obtaining a qualifying education is a key factor for the integration in the labour market for 2^{nd} generation immigrants. However, the economic returns to education as estimated from wage regressions (conditional on employment) are somewhat smaller for 2^{nd} generation immigrants than for ethnic Danes.

The paper is organized as follows. Section 2 presents the hypotheses to be tested in the paper. Section 3 gives an overview of immigration patterns to Denmark, describes the data sources and presents descriptive statistics for the two register-based panel data sets. Section 4 analyses the probability of completing a qualifying education, whereas Sections 5 and 6 analyse the duration of first non-employment and employment spell after leaving the school system. Finally, Section 7 presents estimates of wage functions for the hourly wages in 1997 and Section 8 offers a conclusion and discusses potential policy implications from the analysis of the school-to-work transition of young 2^{nd} generation immigrants.

2. Three main hypotheses

Much of the existing research on 2^{nd} generation immigrants' integration into the labour market has been inspired by the theories and empirical analyses on intergenerational mobility, see for instance the survey by Solon (1999). The theories on intergenerational mobility explain the educational attainment and earnings capacity of the child (or 2^{nd}) generation by the investments and endowment of the parent (or 1^{st}) generation, and other factors influencing the earnings capacity of the children, see Becker and Tomes (1986). A priori, the intergenerational mobility for immigrant families with respect to earnings capacity might be expected to be larger than for native born generations, because the immigrant families tend to assimilate to the host country over time. The assimilation process is often thought of as a 'melting pot' process,² though this hypothesis has been questioned and rejected by a number of empirical studies, indicating that the intergenerational mobility is smaller for immigrants than for natives, see Borjas (1995).

The intergenerational transmissions among immigrants may work through at least three different channels: direct effects from the parents ('parental capital'), effects from the ethnic group ('ethnic capital') and effects from the neighbourhood in which the individual grew up ('neighbourhood effects'). This gives rise to the three main hypotheses in this paper, namely that each of these channels are important determinants when it comes to explaining not only intergenerational mobility, but also labour market entry of young 2nd generation immigrants. As discussed in Solon (1999), it is not straightforward to identify the direct parental skill effects from the ethnicity and neighbourhood effects, since measurement errors in parental skill variables may be captured by the two other effects. Similarly, Ginther, Haveman and Wolfe (2000) find that significant neighbourhood effects tend to disappear when a rich set of family and individual characteristics are included. However, they conclude that the more closely the neighbourhood factor is tied to the outcome under study, the more likely it is to be robust towards addition of covariates. This may be the reason why Borjas (1992, 1995) finds indication that all three effects are important determinants of log earnings and skill levels of American 2nd generation immigrants.

Concerning the first hypothesis - that parental capital is important - there is plenty of empirical evidence that the earnings capacity of the parents is an extremely important determinant of the earnings capacity of the children, see for instance Solon (1999). Here, we extend the analysis and also focus on the potential parental effects on the duration of the first non-employment and employment spells after leaving the school system.

Regarding the second hypothesis - concerning the effect of ethnic capital - Borjas (1992, 1994) introduced the concept as one explanation of lower intergenerational mobility among immigrants compared to natives. The investments in human capital and earnings potential of the immigrant children may be influenced by an external effect from the average human capital of the ethnic group concerned, since originating from a low-skilled ethnic group may reduce the intergenerational mobility. Borjas (1992) labels this an 'ethnic capital' effect, a concept which is fairly parallel to the concept of 'social capital', see Coleman (1988). For immigrants, the notion of ethnic capital may include cultural and linguistic skills which are important for the human

² See Borjas (1999) for an overview.

capital and earnings potential in the host country.

The third hypothesis analysed in intergenerational mobility studies is 'neighbourhood effects'. Traditionally, neighbourhood effects in intergenerational mobility studies are associated with the negative impact of neighbourhoods on children growing up in poor communities. Neighbourhood effects on children's human capital may arise through role modelling, enforcement of social norms by residents of the community and through the influence of social institutions, for instance schools in the neighbourhood, see Solon (1999). The concept of neighbourhood effects has been introduced into the analyses of 2nd generation immigrants by Borjas (1992, 1995), and it may be given a special interpretation when applied to skills and human capital of 2^{nd} generation immigrants. If a child grows up in a community with a high fraction of inhabitants being immigrants, the child is more exposed to meet immigrants and less exposed to meet natives, ceteris paribus. This may have other effects on the skills of the child than the effects of 'ethnic capital'. One obvious neighbourhood effect may arise at school. If the child grows up in a community with a high fraction of school children who are 1st or 2nd generation immigrants, this may influence the proficiency in the language of the host country and more generally the quality of school education. Neighbourhood effects typically refer to childhood experiences. However, in this study we also try to test whether there are any differences between childhood effects and effects arising from the current neighbourhood which may be different if the young individual has moved to another location.

3. Descriptives

3.1. First and second generation immigrants in Denmark

Historically, Denmark has never been an immigration country. Until the mid 1980s, less than 3% of the population were immigrants and the number of 2nd generation immigrants amounted to about only 0.4% of the population. During the last 15 years, these figures have changed considerably. In 2000, 5.6% of the Danish population were 1st generation immigrants and 1.5% were 2nd generation immigrants. Due to the relatively small number of immigrants in the past, the number of adult 2nd generation immigrants is still small and the adult 2nd generation immigrants are on average very young. Out of the 81,000 2nd generation immigrants, only 14,000 were aged 20 years or more in 2000. This is illustrated in Figure 1 which shows the proportion of the population in different age groups for 1st and 2nd generation immigrants.

Figure 1. The number of 1^{st} and 2^{nd} generation immigrants as a proportion of the total population in different age groups in 2000.



Source: Statistikbanken, Statistics Denmark.

Until the early 1970s, the dominant part of immigrants entering Denmark were labour migrants and tied movers. Around 1970 immigration mainly consisted of 'guest workers' from Turkey, Pakistan and the former Yugoslavia, whereas earlier immigrant cohorts consisted almost exclusively of immigrants from developed countries, especially the other Nordic countries (Sweden, Norway, Finland and Iceland). After the first oil crisis in 1973, the immigration pattern changed and as in many other European countries, the immigration of 'guest workers' stopped, while refugee immigration grew considerably, especially after the mid 1980s, see Husted et al. (2001).

Due to these changes in the immigration patterns, the age composition of 2^{nd} generation immigrants varies considerably for immigrants from less developed countries compared to developed countries, see Figure 2. There are relatively few 2^{nd} generation immigrants from less developed countries who are aged more than 25 years while the age composition of 2^{nd} generation immigrants from developed countries is less skewed, relative to the native population.

Figure 2. Second generation immigrants as a proportion of total population in different age groups distributed by country of origin for 2000.



Source: Statistikbanken, Statistics Denmark.

Figure 3. Ethnic origin for 2^{nd} generation immigrants aged 18 to 35 years in 1985 and 1997.



Note: EU-12 are the EU countries, excluding Denmark, Sweden, Finland and Austria. Other developed countries are USA, Canada, New Zealand, Australia, Europe excluding the EU, Norway, Iceland, Japan and parts of the former USSR.

3.2. Data

The empirical analysis is based on two large data sets, originating from administrative registers in Statistics Denmark. The first data set contains information on the entire population of immigrants in Denmark (about 270,000 individuals in 1997) for each of the years 1985-1997. The other data set contains 10% of the Danish population (about 500,000 individuals) covering the period 1985-1997. The original sample is an unbalanced panel sample of individuals aged 15 years or more. Young individuals are added to the sample each year. Thus, besides being a panel, the sample is also representative for the population in each of the years. The samples contain information on a large number of demographic and labour market characteristics of the individuals and their families.

In this study, the two master samples are restricted to individuals aged 18-35 years. From the immigrant database, we select all 2nd generation immigrants, i.e. individuals who are born in Denmark and who have 1st generation immigrant parents.³ In 1997, there were 4,579 men and 4,345 women in this group. Because the database contains the total population of 1st and 2nd generation immigrants, it is possible to identify a number of characteristics of 2nd generation immigrants' parents. The information used about the parents is: years since migration, country of origin, the municipality and county where they live (Denmark has 276 municipalities and 13 counties), the amount of education⁴, and the annual gross income. A weakness of the database used is that it does not include information on Danish language proficiency of the parents. For the sample of young natives, which includes about 66,000 men and 64,000 women in 1997, the same information is available, allowing us to compare the intergenerational transmission pattern of immigrants to that of the natives.⁵

The three main hypotheses in this paper regard the effect of parental capital, ethnic capital and neighbourhoods. As concluded in Österberg (2000), it is not sufficient to include, for instance, education of the parents when the education of the child is studied. One reason is that many

³ We use the official definition used by Statistics Denmark of a 1^{st} generation immigrant: A first generation immigrant is defined as an individual with birth country other than Denmark where both parents have foreign citizenship or are born abroad. If information on one of the parents is missing but the other parent is full-filling the criteria, the individual is also defined as an immigrant. Finally, if there is no information on any of the parents then the individual is defined as a first generation immigrant if he is born abroad.

⁴ Information on parents' education acquired before migration is available for immigrants present in 1999, where questionnaires were sent to the entire population of 1st generation immigrants. The issue of education of 1st generation immigrants is discussed in Mørkeberg (2000).

⁵ The sample of ethnic Danes only contains information about the parents for individuals who were aged less than 20 years in 1980. Therefore, we only use individuals who fulfill this criteria.

immigrants have a lower economic position than equally educated natives do. To represent parental capital we use: Number of years since migration, length of education, the log annual gross income and labour market experience in Denmark. We use the maximum values of the variable for the father and mother, if information on both parents exists.⁶ If there is only information on one of the parents, we use this information. For a considerable number of 2nd generation immigrants (5-6%), we do not have any information on their parents, either because both parents are dead or have emigrated from Denmark. Therefore, we include an indicator variable for missing information.

In order to analyse the effects of ethnic capital, we have constructed a measure which is the average value for 1st generation immigrants in the ethnic group concerned of each of the four variables included in parental capital described above, i.e. years since migration, length of education, annual income and working experience obtained in Denmark. The ethnic group is defined by country of origin and for each country of origin, we select the larger of the average for males and females.

The variables measuring neighbourhood effects consist of four variables, two for the childhood neighbourhood and two for the current neighbourhood: The database includes information on the municipality in which the individuals (and the parents) live during each year in the whole observation period. Based on this information, we construct a variable which we call 'ethnic concentration' defined as the percentage of 1st and 2nd generation immigrants in the population in the municipality concerned. A supplementary variable, an 'exposure index' (see Borjas, 1999), is defined as the proportion of immigrants (1st or 2nd) with a certain ethnic origin in the municipality divided by the corresponding proportion in Denmark as a whole. This variable gives an indication of the relative density of immigrants from own country of origin which characterizes the municipality in which the individual lives. A negative effect of this variable might be interpreted as a 'ghetto effect', whereas a positive effect might be interpreted as an 'ethnic good' or 'ethnic network' effect.

⁶ One might argue that the characteristics of both parents should enter as explanatory variables. However, this would dramatically increase the problem of missing information on the variables of the parents. But since we only include the values of one of the parents it may give rise to problems when comparing natives and immigrants with respect to intergenerational effects, see Card et al. (1998). If the parents of immigrants are more likely to have similar characteristics (income and education) than ethnic Danish parents, and if both the characteristics of fathers and mothers affect the outcome of the children, we tend to get higher estimates of the effects of parental capital for immigrants than for ethnic Danes.

These two variables are constructed for both the childhood neighbourhood, which is of main interest in the context of intergenerational transmission, but also for the current neighbourhood, that is where the individual resides at the time of observation to capture potential assimilation effects. The childhood neighbourhood is defined as the neighbourhood of the parents in 1985, the first year of observation in our sample. For the younger part of the individuals in our sample this measure will catch the childhood neighbourhood fairly well but for the oldest groups who might already have left their parents' home in 1985, this measure is only correct to the extent that the parents did not change their residence since the children were young and lived in their households.⁷ We expect that 'childhood neighbourhood' effects capture the influence of role models as well as for instance the effects of school quality, while the 'current neighbourhood' effects capture two intuitively counteracting effects, namely a (positive) network effect and a (negative) ghetto-effect. However, there may be endogeneity problems with these variables which we do not account for in this study since the location of immigrants, both 1st and 2nd generation, may in itself be affected by the preferences and the labour market prospects of the immigrants.⁸

In Table A1, the raw means of the neighbourhood variables are shown, whereas the variation across ethnic origin is seen in Table A2. Second generation immigrants from Pakistan tend to live in municipalities with many immigrants, and they also tend to live in areas where the relative frequency of immigrants from Pakistan is very high, as indicated by the 'exposure index'. Second generation immigrants originating from other less developed countries also typically live in municipalities with a high fraction of immigrants, but the exposure index is on average much lower than for Pakistani immigrants. Looking at the same variables for the parents (not shown in Table A2), exactly the same pattern appears for the parents of the 2nd generation immigrants, indicating that the children could be settling in the same geographical areas as their parents.⁹

Finally, we use a number of variables which typically enter human capital models or models explaining employment and educational attainment. Age is represented by a number of indicators for age cohorts, education is measured by length of education, and the experience variables (experience and experience squared) measure number of years of actual experience obtained by

⁷ The municipality of the mother determines the measure of the variable 'ethnic concentration, childhood municipality' because the children most often live with their mother in case the parents have divorced. If information on the mother is missing (typically if the mother is dead) we use the information of the father.

⁸ Edin, Frederiksson and Åslund (2000) present Swedish evidence.of sorting across locations.

⁹ This is confirmed in the study by Hummelgård et al. (1998).

being employed as a wage earner in Denmark.¹⁰ We also include a marriage indicator and for immigrants, we distinguish between whether the spouse is an immigrant or a native (the excluded category consists of unmarried cohabitants and singles). These variables are included to indicate the existence of a Danish network. However, these variables may suffer from endogeneity problems because the ethnic origin of the spouse may be influenced by, among other factors, the educational attainment of the individual.

The databases refer to the period 1985-97 which is a period of very large economic fluctuations in Denmark, including strong peaks in 1985-86 and 1996-97 and a deep recession in 1991-93. Furthermore, there are fairly large variations in regional unemployment rates during the period, with the highest unemployment rates in the Copenhagen region for most of the period. In order to capture these cyclical and regional effects, we include a variable for the unemployment rate in the local labour market in the models of duration of non-employment and first employment spell.¹¹

We have estimated four alternative specifications for 2nd generation immigrants where more explanatory variables successively are added: (i) A model including parental capital variables beside individual variables for the 2nd generation immigrants, (ii) to this model we add ethnic capital variables, (iii) and further the model is extended with neighbourhood variables (only childhood neighbourhood). Finally, in (iv) we include current neighbourhood variables and a number of indicators for country of origin and region in Denmark, mainly in order to test the sensitivity of the coefficients of ethnic capital and neighbourhood variables, ethnic capital and neighbourhood effects turn out to be very robust with respect to these four specifications and

¹⁰ This variable is based on the payments to a compulsory pension scheme, ATP. These payments are registered for all individuals employed as wage earners in Denmark back to 1964. Based on this information, we are able to construct a rather precise measure of accumulated experience. The deficiency is that until recently ATP has not been paid by self-employed individuals or assisting spouses and thus, these activities do not add to the accumulated experience. Since many immigrants are self-employed, we expect to underestimate the actual work experience of immigrants.

¹¹ We define the local labour market for each individual as the region within which the individual can commute at an average transportation cost of DKK 90. This variable is originally constructed by the Institute of Local Government Studies (AKF) who uses it in their regional models. AKF has kindly supplied us with the information necessary to construct the variable. The variable is treated as time-invariant since it measures the unemployment rate at the beginning of the spell.

¹² The indicators for country of origin are grouped into Turkey, Pakistan, other less developed countries, Nordic countries, EU-12 and other developed countries. The regional indicators are based on the 13 Danish counties. At an earlier stage, we also included an indicator for whether the individual originated from a refugee family, but that indicator had no explanatory power.

therefore, we only show the results from the most extended model (iv). In order to compare with Danes, we have estimated the same model (i.e. model iv) for young ethnic Danes, too. For young ethnic Danes we do not include the variable 'ethnic capital' and the exposure index. These variables are not meaningful for the group of ethnic Danes because they are either constant or show very little sample variation.

3.3. Descriptive statistics for young natives, 1st and 2nd generation immigrants

In Table 1, we compare some key variables for 1^{st} generation immigrants, 2^{nd} generation immigrants and ethnic Danes in 1985 and 1997. Even though the age group has been restricted to 18-35 years, it is clear from Table 1 that the group of 2^{nd} generation immigrants is younger in 1997 than it was in 1985 because the young age cohorts passing the age of 18 during this period are much larger than the older age cohorts of 2^{nd} generation immigrants.

(Values for males above, for	r females bel	ow).				
	1985				1997	
	Ethnic Danes	1 st generation immigrants	2 nd generation immigrants	Ethnic Danes	1 st generation immigrants	2 nd generation immigrants
Age	26.2	27.6	25.5	26.8	28.1	23.3
	26.2	27.6	25.1	26.8	27.4	23.3
Education, years	11.2	5.5	11.1	11.7	5.7	10.8
	11.2	5.6	11.3	11.9	5.7	11.0
Experience, years	6.5	3.8	4.9	7.4	2.3	3.3
	5.5	3.1	4.0	5.7	1.5	2.7
% Employed ¹⁾	69	45	51	64	33	33
	57	33	44	48	20	29
Individual unemployment rate	9.4	14.7	10.2	6.8	15.1	9.5
	14.9	16.8	13.9	10.0	12.5	9.4
Hourly wages, DKK 1997 prices ²⁾	138.6	139.9	140.9	156.3	144.4	153.9
	110.4	116.1	114.9	128.0	124.6	128.2

Table 1. Mean values of key variables for ethnic Danes, 1^{st} and 2^{nd} generation immigrants aged 18-35 years. (Values for males above, for females below).

1) Employed at least 1000 hours during the year as paid worker or registered as self-employed. The employment shall be on ordinary market conditions, i.e. not employment in a labour market programme.

178.3

133.7

2,221

1,903

202.6

154.1

65,715

63,507

139.5

103.6

48,111

46,458

139.7

120.0

4,498

4,259

2) The calculation of wage rates is restricted to individuals who are employed and not enrolled in the educational system.

144.1

100.9

24,386

23,134

3) Danes 10% and immigrants 100% of the population.

140.3

100.6

69,321

66,155

Annual gross income,

thousand DKK 1997 prices

Sample/population size 3)

The most striking fact from Table 1 is that 2nd generation immigrants are much better off compared to 1st generation immigrants when looking at their labour market characteristics - even though they are much younger on average. In general, this holds true for all key variables reported in both 1985 and 1997: Education, experience, employment and unemployment rate, hourly wage and annual gross income.¹³ However, in 1997 the employment rates of male 1st and 2nd generation immigrants are equal. And, in 1985 the hourly wage for female 2nd generation immigrants is slightly lower than that of their peers from the 1st generation.

Comparing 2nd generation immigrants to ethnic Danes, the labour market performance is generally much better for ethnic Danes due to the different age compositions of the two groups, at least in 1997. However, when it comes to hourly wages and annual gross income, 2nd generation immigrants rank first in 1985 whereas ethnic Danes rank first in 1997. In addition to the declining average age of 2nd generation immigrants, the explanation is found in the changing composition by ethnic origin during the period. In 1985, individuals originating from Nordic and EU-12 countries dominated the population of 2nd generation immigrants and they tend to be older and earn high wages, whereas in 1997, individuals originating from less developed countries dominated and they tend to be younger and have less favourable labour market characteristics (see Table A2 in the appendix).

Table 1 shows only minor differences in the hourly wage rates between ethnic Danes and 1st and 2nd generation immigrants in both 1985 and 1997. It is surprising as one would expect the large differences in observed human capital to result in larger wage differentials. However, as argued by Husted et al. (2001), it may have to do with the compressed wage structure in Denmark implying that differences in human capital tend to show up in terms of different employment rates and consequently different gross incomes, as also indicated by Table 1. In the Appendix, Table A2 shows a more differentiated picture of the wage and income differentials across ethnic origin. Here the differences in mean age and education clearly show up in the hourly wage and the annual gross income. However, 2nd generation immigrants with ethnic origin in the Nordic and EU-12 countries earn higher hourly wages than others including ethnic Danes, although the age and level of education are similar.

¹³ This was also found by Hummelgård et al. (1998), and Ekberg (1997) illustrates a similar picture for Sweden.

Figure 4. Fraction of age cohort of 2^{nd} generation immigrants and ethnic Danes who have completed a qualifying education in 1997.



Note: See note to Figure 3 about the country grouping.

4. Educational attainment of 2nd generation immigrants

It is usually thought that a basic condition for a successful integration in the labour market is to obtain a qualifying education. In this study, we focus on whether or not a qualifying education has been obtained, and not on different types or lengths of educations.¹⁴ Figure 4 shows the proportion of each age cohort who has completed a qualifying education in 1997. For Turkey, Pakistan and 'Other less developed countries' the figure has been cut off at the age of 24 because there are only few observations in older age cohorts.

The general impression from Figure 4 is that 2nd generation immigrants from EU, the Nordic countries and other developed countries do not deviate much from ethnic Danes with respect to the probability of completing a qualifying education. For 2nd generation immigrants from less developed countries, the story is different. Among 2nd generation immigrants from Turkey, a much lower fraction in each age cohort has completed a qualifying education. For Pakistan and other less developed countries the difference to ethnic Danes exists although it is smaller. The figure shows that female 2nd generation immigrants from Turkey, Pakistan and other less developed countries do fairly well in the educational system compared to males, although they still lack slightly behind their native peers (see also Table A2).¹⁵

¹⁴ Only an education lasting at least 18 months (beyond secondary school and/or highschool) is considered a qualifying education.

¹⁵ Since the individuals included in Figure 4 are young on average, a possible hypothesis for the low fraction among individuals with Turkish origin is that they are still undertaking education. For Turkish men, detailed analyses reject this hypothesis, since a larger proportion of this group compared to others immigrants and natives is neither under education, nor have they completed a qualifying education. For Turkish women, that tendency is less pronounced.







Despite the relatively low level of education for male Turkish 2nd generation immigrants, this group has a much higher level of education than their parents. The same holds true for many other 2nd generation immigrants, as illustrated in Figure 5, which shows the average level of education for parents, sons and daughters for the 108 countries of origin registered in the administrative registers.¹⁶ There is a large variation in the educational attainment of both parents and their children between the 108 countries and in general, children have obtained a much higher level of education than their parents. Based on the figures, it seems fair to conclude that there is a positive correlation between the level of education for parents and their children, both sons and daughters. This correlation may stem from influence directly from the parents or from the ethnic group concerned. However, the level of education for the children may also depend on many other determinants than the education of the parent generation. In order to analyse this further, we estimate a model for having completed a qualifying education. Since we only focus on a dichotomous measure of education, we use a simple logistic regression. Let F denote the logistic CDF, let P_i , E_i , and N_i denote the three sets of explanatory variables related to intergenerational transmissions, namely, parental capital, ethnic capital and neighbourhood variables, and let O_i denote other variables. Finally, let D_i be an indicator for obtaining a qualifying education, then the likelihood contribution for individual *i* is given by:

(1)
$$\ell_i(.) = F(\alpha X_i)^{D_i} \cdot (1 - F(\alpha X_i))^{1 - D_i}, \text{ where } X_i = \{P_i, E_i, N_i, O_i\}$$

As mentioned earlier, the intergenerational transmission from ethnic capital (E_i) is not relevant for young natives but only for individuals with a non-Danish ethnic origin. The matrix O_i includes marriage indicators, age indicators, regional indicators and indicators for missing information on parents. The results from maximum likelihood estimation are shown in Table 2.

¹⁶ The graph for the relationship between mother and son is not presented because this is where the lowest correlation is usually found, see e.g. Österberg (2000).

	Dep. variable = 1, if finished qualifying education, 0 otherwise				
	Mal	es	Fema	lles	
	2 nd generation immigrant	Ethnic Danes	2 nd generation immigrant	Ethnic Danes	
Parental variables: Years since migration parents/100 ¹⁾	2.158* (0.755)	-	0.697 (0.729)	-	
Education, years, parents/100 ¹⁾	0.335 (1.013)	4.517* (0.220)	-0.707 (1.020)	3.798* (0.224)	
Log annual gross income, parents/100 ¹⁾	2.673 (3.625)	-0.737 (0.592)	7.669* (3.880)	-0.068 (0.563)	
Experience, years, parents/100 ¹⁾	2.083* (0.541)	2.736* (0.140)	1.590* (0.546)	2.157* (0.140)	
<i>Ethnic capital variables</i> : Average years since migration	0.091 (0.078)	-	-0.006 (0.075)	-	
Average education ²⁾	-0.031 (0.054)	-	-0.002 (0.055)	-	
Average log annual gross income	0.002 (0.629)	-	0.005 (0.652)	-	
Average years of experience	-0.002 (0.055)	-	-0.030 (0.054)	-	
<i>Neighbourhood variables:</i> Ethnic concentration in childhood municipality ³⁾	-8.094* (1.504)	-5.894* (0.432)	-3.418* (1.495)	-6.136* (0.437)	
Exposure index, childhood municipality 4)	0.060* (0.024)	-	0.005 (0.024)	-	
Other variables: Constant	-3.534 (6.800)	-1.266* (0.076)	-2.000 (7.080)	-1.521* (0.074)	
Married to an ethnic Dane ⁵⁾	0.341* (0.140)	0.495* (0.026)	0.296* (0.136)	0.357* (0.023)	
Married to an immigrant	0.056 (0.412)	-	-0.583* (0.248)	-	
Indicators for age	yes	yes	yes	yes	
Indicators for region of origin	yes	-	yes	-	
Indicators for Danish regions	yes	yes	yes	yes	
Indicators for missing information on parents	yes	yes	yes	yes	
Mean log likelihood	-0.593	-0.615	-0.619	-0.618	
No. of observations	3186	58420	2998	56684	

Table 2. Logit model for having completed a qualifying education in 1997, individuals aged 20-35 years.

* Significant at a 5% significance level.

1) Defined as the maximum values of either father's or mother's variable. If information on both parents is missing, the value is set to 0, and in that case the indicator for missing information on parents take the value of 1.

2) Ethnic capital variables are defined as the country average for 1^{st} generation immigrants for the variable concerned.

3) Ethnic concentration is defined as the proportion of 1^{st} and 2^{nd} generation immigrants in the municipality concerned.

4) Exposure index is defined as the fraction of immigrants from own ethnic origin in the municipality concerned divided by the fraction in the total population from own ethnic origin.

5) For ethnic Danes this indicator is equal to unity whenever the individual is married.

The completion of a qualifying education seems to be highly affected by intergenerational transmission effects working through parental capital and childhood neighbourhoods. For male 2nd generation immigrants, the number of years since the parents immigrated to Denmark has a positive effect on the probability of completing a qualifying education, whereas for women, this effect is insignificant. On the other hand, the education of young 2nd generation immigrant women seems to be highly dependent on the income the parents. This is not the case for young immigrant men - or for ethnic Danes. Having parents who have several years of labour market experience from the Danish labour market has a significantly positive effect on the probability of completing a qualifying education of the parents is only significantly positive for young Danes. This is quite surprising, and it suggests that there is substantial intergenerational mobility among the immigrants, since this is the variable which is most closely related to the outcome variable under study. It may also be that there are measurement errors in the variable indicating the educational attainment of 1st generation immigrants.

We are not able to identify any significant ethnic capital effects in this model (having controlled for region of origin). It could be argued that the four variables representing ethnic capital are partly collinear but other estimations show that this result also holds if only one or two ethnic capital variables are included. Turning to the neighbourhood effects, the estimations yield some interesting results. For all groups, immigrants as well as young Danes, men as well as women, growing up in an area with a high concentration of immigrants is associated with a significantly lower probability of completing an education. The size of the effect is surprisingly large although, as mentioned above, the estimated coefficients may suffer from endogeneity. Evaluated at the sample mean, the marginal effect of increasing ethnic concentration from 10% to 20% (corresponding to moving from a municipality with just above average level ethnic concentration to one of the Danish municipalities with the highest ethnic concentration) is associated with a reduction in the probability of completing a qualifying education with 18 percentage points for a 2nd generation young man and 13 percentage points for a young ethnic Danish man. For women, the negative effects are slightly smaller, 8 percentage points and 10 percentage points for immigrants and natives, respectively. On the other hand, living in an area with a relatively large fraction of immigrants with the same ethnic origin is associated with a significantly higher probability of obtaining a qualifying education for 2nd generation immigrant men (but not women). This may reflect that young immigrant men are more dependent on a social network and that the concept of 'ethnic goods' is more important for young immigrant men than women.

Finally, marriage matters for the attainment of education. Being married to an ethnic Dane is associated with a higher probability of completing a qualifying education. The marginal effect is about 7 percentage points for an average individual. For an average 2nd generation immigrant women, marriage to an immigrant man reduces the probability of completing a qualifying education by about 14 percentage points. Although one should be careful with the interpretation because the causality may run in both directions, this negative effect is a serious drawback for young female immigrants from less developed countries since only about 2% of these women marry an ethnic Dane (Pedersen, 2000), and among Turks and Pakistanis three-fourths of all immigrants are against marriage with an individual of a different ethnic origin (Nielsen, 2000).

5. The duration of waiting time from leaving the educational system until the first job

After having left the educational system with or without completing a qualifying education, the next question is how long time, the young 2^{nd} generation immigrant or native has to wait until he or she succeeds in finding a job in the ordinary labour market.¹⁷ In the second model, we estimate the duration (in months) until the first job in the ordinary labour market after leaving the educational system. In Table A1, the average length of this waiting time (non-employment spells) is shown. The average waiting time is longer for 2^{nd} generation immigrants than it is for ethnic Danes and in addition, it is longer for women than it is for men.

We estimate a proportional hazard model with a piecewise constant baseline hazard. If t_i denotes the waiting time for the first job, the hazard function looks as follows:

(2)
$$h(t_i | X_i) = h_0(t_i) \exp(\beta X_i)$$
, where $X_i = \{P_i, E_i, N_i, O_i\}$

To the vector O_i is now added years of education and the regional unemployment rate as earlier described. The splitting times for the baseline have been determined by looking at Kaplan-Meier hazard plots. A large group of individuals finds employment almost immediately after leaving school, hence we have chosen a finer grid at the beginning of the spell. The splitting times chosen for both men and women, 2^{nd} generation immigrants and ethnic Danes are: 1,3,6, 12, and 24 months.

¹⁷ We do not consider participation in different labour market schemes (publicly subsidized jobs, job offer schemes or in-work-benefits) as ordinary jobs.

	Estimates are effects on the log hazard rate for entering the first job				
	Mal	es	Fema	lles	
	2 nd generation immigrant	Ethnic Danes	2 nd generation immigrant	Ethnic Danes	
Parental variables:	0.758*	_	1.597*	-	
Years since migration, parents/100 ⁻¹	(0.263)		(0.274)		
Education, years, parents/100 ¹⁾	0.315 (0.419)	-1.148* (0.094)	-1.267* (0.462)	-0.690* (0.113)	
Log annual gross income	0 311	3 483*	3 714	3 719*	
parents/100 ¹⁾	(1.497)	(0.531)	(2.247)	(0.577)	
Experience, years, parents/100 ¹⁾	0.278	1.656*	1.379*	0.828*	
	(0.270)	(0.061)	(0.304)	(0.069)	
Ethnic capital variables:	0.010		0.017		
Average years since migration	-0.012 (0.026)	-	-0.046 (0.037)	-	
Average education ²⁾	-0.020	-	-0.073*	-	
-	(0.015)		(0.030)		
Average log annual gross income	0.201 (625.920)	-	-0.019 (0.311)	-	
Average years of experience	0.075* (0.025)	-	0.017 (0.026)	-	
Neighbourhood variables:					
Ethnic concentration in childhood municipality ³⁾	-3.228* (0.834)	-2.269* (0.217)	1.274 (1.005)	-0.846* (0.221)	
Exposure index, childhood municipality 4)	0.002 (0.016)	-	0.019 (0.018)	-	
Ethnic concentration in current municipality ³⁾	0.018 (0.744)	-2.237* (0.165)	-1.447 (0.930)	-1.962* (0.168)	
Exposure index, current municipality ⁴⁾	-0.053* (0.015)	-	-0.035* (0.018)	-	
Other variables:					
Education, years	0.066*	0.059*	0.092*	0.073*	
	(0.008)	(0.002)	(0.009)	(0.002)	
Regional unemployment rate	-0.107* (0.011)	-0.060* (0.002)	-0.089* (0.011)	-0.055* (0.002)	
Married to an ethnic Dane $5^{(1)}$	0 222*	0.438*	0.122	0.064*	
Married to an entitle Dane	(0.089)	(0.023)	(0.107)	(0.018)	
Married to an immigrant	0.427 (0.249)	-	-0.664* (0.232)	-	
Piece-wise linear baseline hazard	yes	yes	yes	yes	
Indicators for age	yes	yes	yes	yes	
Indicators for region of origin	yes	-	yes	-	
Indicators for Danish regions	yes	yes	yes	yes	
Indicators for missing information on parents	yes	yes	yes	yes	
- Mean log likelihood	-1.153	-0.512	-1.224	-0.922	
No. of observations	1633	20403	1582	21629	

Table 3. Duration model for time until first ordinary job, individuals aged 18-35 years.

Notes: See Table 2.

Looking at the baseline hazard rates (which are not shown in Table 3), we find that the hazard rate is very high during the first month after leaving school, reflecting the fact that many school leavers find jobs immediately. After the first month, the hazard rate tends to be constant until 12 months after leaving school after which point it declines. For 2nd generation immigrant women, there is negative duration dependence throughout all baseline intervals. This negative duration dependence might indicate that the individuals who have not found job after some time are negatively selected with respect to unobserved characteristics (e.g. motivation or language proficiency), that is, the negative duration dependence may be spurious, see Lancaster (1990). On the other hand, it may be genuine negative duration dependence, in which case it may reflect stigma effects of non-employment, depreciating human capital, etc.

Focusing on the intergenerational transmission mechanisms, the dominant effect for women comes from the parents, whereas men are affected by ethnic capital and childhood neighbourhoods as well. For natives, we find significant effects of both parental capital and neighbourhood variables.

Spelling out the results in more detail, the coefficients on the 'parental capital' variables (years since migration, log annual gross income and labour market experience) confirm that assimilation and labour force attachment of the parents have a significantly positive effect for mainly female 2nd generation immigrants, but these effects are not as well-determined for males for whom it is only the years since migration of the parents that matters. The striking gender difference in the strength of the effect of parental capital may indicate the existence of 'parental gender discrimination' when a girl child considers entering the labour market. This 'discrimination' may stem from inducing the girls to getting married (with a man of similar nationality) and being a full-time housewife instead of entering the labour market.

For female 2nd generation immigrants as well as ethnic Danes, the education of the parents seems to exert a 'perverse' effect on the probability of entering the first job after leaving the educational system. For some reason, a higher education of the parents (or their generation) prolongs the non-employment spell. However, compared to the effect of own education (see below), the effect is relatively small.

When it comes to the males, it is important whether they belong to an ethnic group who has been attached to the Danish labour market for many years. This may indicate that the integration process of male 2nd generation immigrants is highly dependent on the labour force integration of the parent generation, and belonging to an ethnic minority who has spent many years in Denmark without accumulating labour market experience has negative effects on the integration process.

Furthermore, the labour market success of male 2^{nd} generation immigrants is affected by the concentration of immigrants in the neighbourhood where they grew up, as it is the case for natives of both gender. For natives, but not for immigrants, the negative ethnic concentration effect is reinforced if the individual currently lives in an ethnic concentrated municipality. Thus, young 2^{nd} generation immigrant women do not suffer from coming from an ethnically concentrated neighbourhood, while this is a clear disadvantage for all other groups. Regarding the exposure index, we find that the overall effect on the job finding rate of living in a municipality with an above normal share of immigrants from the same ethnic group is negative.

As expected, the level of own education for the 2nd generation immigrant is extremely important. Individuals who have completed a long education are likely to find a job faster than others. Thus, having a qualifying education significantly increases the probability of obtaining employment for all groups, a result which is in line with earlier Danish studies of 1st generation immigrant men, see Husted et al. (2001). Those who leave the educational system when the unemployment rate is high tend to spend significantly longer time finding a job. Finally, a married young immigrant man has a shorter waiting time until the first job, whether married to another immigrant or to an ethnic Dane. On the contrary, for young immigrant women, the effect of being married to another immigrant is strong and significantly negative.

6. Duration of the first employment spell

We analyse the duration of the first spell of employment, given that the immigrant succeeds in obtaining a job.¹⁸ We are not able to get exact information on the duration of jobs but based on the administrative registers, we are able to calculate the duration (in months) of the first employment spell, which may consist of several job spells. Only jobs in the ordinary non-subsidized labour market are included. The average length of the first employment spell is longer for ethnic Danes than it is for 2nd generation immigrants (see Table A1). Once again, the estimated model is a proportional hazard model with a piecewise constant baseline. Let s_i denote the duration of the employment spell for individual *i*:

(3)
$$h(s_i | X_i) = h_0(s_i) \exp(\theta X_i)$$
, where $X_i = \{P_i, E_i, N_i, O_i\}$

Choice of baseline intervals (or equivalently splitting times) is based upon an examination of Kaplan-Meier hazard plots (not presented here), and the following splitting times are chosen: 3, 6, 9, 12 and 24 months.

¹⁸ Ending employment is equivalent to becoming unemployed in this study, as we treat all other destination states out of employment as independent right censoring.

In general, there is a negative duration dependence, although there is a tendency that the hazard rate is higher during the third to sixth month of employment for 2nd generation immigrants, as well as for ethnic Danish males, while for ethnic Danish females, the hazard rate out of employment is higher during the seventh to twelfth month of employment. The hump-shaped baseline hazard could be explained by 'learning about the quality of the job match' as argued by Jovanovic (1979).

Given that the young immigrant succeeds in getting a job, the effect of intergenerational transmission on the duration of the first employment spell seems to be less significant than on obtaining a qualifying education and on the waiting time until the first job. It might be argued, of course, that intergenerational effects are already captured by the variable reflecting educational attainment, and that there is a strong indirect intergenerational effect present.

For male 2nd generation immigrants, both parental and ethnic labour market experience reduce the instantaneous probability of leaving employment (significant at a 10% level). This effect is also found for ethnic Danes but not for female 2nd generation immigrants. For female 2nd generation immigrants, on the other hand, we find that after controlling for different background variables (e.g. region of origin), there is a positive effect (significant at a 10% level) of the number of years that the parents have been in Denmark. This means that the longer parents have been in Denmark, the shorter is the duration of a female employment spell. Altogether, the results indicate some influence on employment duration of the parental generation, though it is not as well-determined as at the earlier stages of the school-to-work transition. The results add to the evidence presented earlier that integration into the Danish labour market is far from being an 'automatic process' where it is just a matter of time, years or decades, until the immigrants or their children obtain a well-established position on the labour market.

	Male	es	Females		
	2 nd generation immigrant	Ethnic Danes	2 nd generation immigrant	Ethnic Danes	
Parental variables: Years since migration, parents/100 ¹⁾	-1.328 (0.864)	-	1.875* (0.946)	-	
Education, years, parents/100 ¹⁾	0.299 (0.818)	0.065 (0.269)	0.131 (0.895)	0.427 (0.268)	
Log annual gross income, parents/100 ¹⁾	2.971 (3.752)	-2.402* (0.986)	2.272 (4.781)	-0.016 (1.069)	
Experience, years, parents/100 ¹⁾	-0.955 (0.553)	-1.001* (0.160)	-0.132 (0.606)	-0.951* (0.156)	
<i>Ethnic capital variables:</i> Average years since migration/10	-0.332 (0.509)	-	0.388 (0.715)	-	
Average education/100 ²⁾	-1.220 (3.125)	-	-5.154 (6.235)	-	
Log annual gross income/100	-1.440 (106.416)	-	-2.308 (65.985)	-	
Average years of experience/100	-9.582 (5.221)	-	-3.272 (5.035)	-	
<i>Neighbourhood variables</i> : Ethnic concentration in childhood municipality ³⁾	6.227* (1.951)	2.669* (0.568)	-2.458 (2.065)	0.929 (0.526)	
Exposure index, childhood municipality 4)	-0.050 (0.035)	-	0.067 (0.035)	-	
Ethnic concentration in current municipality ³⁾	-3.559* (1.777)	-0.173 (0.449)	3.720 (1.901)	0.155 (0.413)	
Exposure index, current municipality ⁴⁾	0.077* (0.035)	-	-0.058 (0.036)	-	
Other variables: Education, years	-0.039* (0.015)	-0.005 (0.005)	-0.083* (0.017)	-0.073* (0.005)	
Regional unemployment rate	0.020 (0.022)	0.042* (0.005)	0.003 (0.024)	0.016* (0.005)	
Married to an ethnic Dane ⁵⁾	-0.150 (0.193)	-0.332* (0.068)	-0.366 (0.206)	-0.204* (0.043)	
Married to an immigrant	-0.597 (0.573)	-	0.212 (0.363)	-	
Piece-wise linear baseline hazard	yes	yes	yes	yes	
Indicators for age	yes	yes	yes	yes	
Indicators for region of origin	yes	-	yes	-	
Indicators for Danish regions	yes	yes	yes	yes	
Indicators for missing information on parents	yes	yes	yes	yes	
Mean log likelihood	-2.17	-2.184	-2.135	-2.22	
No. of observations	1516	19265	1356	19704	

Table 4. Duration model for first employment spell, individuals aged 18-35 years.

Estimates are effects on the log hazard rate for ending first employment spell

Notes: See Table 2.

A high ethnic concentration in the childhood neighbourhood of young men, immigrants as well as ethnic Danes, increases the risk of becoming unemployed, but for immigrants the effect is, to some extent, counteracted by the effect of ethnic concentration in the municipality of current residence. For 2nd generation immigrant women, the effects are similar, but insignificant. The exposure indices for the childhood or current municipality also tend to (net) increase the risk of losing the job. Thus, we do not find significant indication of positive 'ethnic network' or 'ethnic good' effects on the chances of keeping a job for young 2nd generation immigrants in Denmark.¹⁹

A longer own education is associated with a lower probability of becoming unemployed after the first employment. Especially for women, this effect is very important. For immigrants, the coefficients to the marriage indicators are not well-determined, though female immigrants tend to be less likely to loose their jobs if they are married to ethnic Danes. Surprisingly, the duration of employment for immigrants who succeed in getting a job is less sensitive to local unemployment rates compared to young Danes. The coefficient to the local unemployment rate is significantly positive for Danes, but insignificant for male as well as female immigrants. This seems to indicate a fundamental difference between the jobs that 2nd generation immigrants and young ethnic Danes get. However, this question is left for future study.

7. Hourly wages in the first employment spell

Finally, the fourth model explains the wage in the first employment spell. The dependent variable is the log hourly wage rate, which is measured in DKK. The information on wages is based on annual earnings divided by annual hours employed.²⁰ Hourly wages are only observed for individuals who have been employed as wage earners during the year while there is no information on the earnings capacity of self-employed individuals who are excluded from this analysis. The means of the hourly wage rates are shown in Tables A1-2. It is higher for males than for females, and higher for ethnic Danes than for immigrants, except for 2nd generation immigrants from the EU and the Nordic countries who have higher hourly wage rates than ethnic Danes. The plots of wage distributions show a log-normal shaped distribution, see Figure 6, but there is no clear indication of large differences with respect to variance or means for the groups included in the figure.

¹⁹ In alternative estimations (not shown here), we have experimented with different interaction terms on the neighbourhood variables, allowing for different effects of these variables in big cities and rural areas. One hypothesis could be that living in a small municipality with a high ethnic concentration would have smaller effects than living in a large municipality with the same high ethnic concentration. However, we found no significant differences in the neighbourhood effects.

²⁰ Thus, overtime payments and earnings in a secondary job are included in the wage measure. If the frequency of overtime work or secondary jobs varies systematically between immigrants and native-born, we may over-or underestimate the differences between the wage levels of immigrants and native-born individuals.

We perform simple human capital estimations of hourly wage rates for young ethnic Danes and 2^{nd} generation immigrants who are observed to have at least one employment spell during the observation period 1985-1997.²¹ The wage equation takes the following form

(4)
$$\ln w_i = \delta X_i + \varepsilon_i, \text{ where } X_i = \{P_i, E_i, N_i, O_i\}$$

To the matrix O_i are now added: years of education, experience and experience squared. Assuming $\varepsilon_i \sim iid(0, \sigma^2)$, the equation is estimated using OLS.

The results are presented in Table 5. While parental capital work as an important intergenerational transmission channel in obtaining education and employment for young 2^{nd} generation immigrants, the hourly wage rates for those who succeed in getting a job do not seem to depend on these variables. And, for natives, there is a perverse effect of parental capital, which - in a society with a highly compressed wage distribution²² - could be the result of a 'regression towards the mean', i.e. those whose parents earn a lot tend to earn a little less, and vice versa. Combined with the results in Table 4, where the effect of these variables also tend to be weak, we may interpret this result as reflecting that parental capital mainly affects the beginning of the school-to-work transition, i.e. the educational attainment and entry into the first job. The direct effects of the background of the parents on the wage rate are weak or non-existing. However, for female 2^{nd} generation immigrants, there is some effect from ethnic capital variables and for male 2^{nd} generation immigrants, there is some effect from the childhood neighbourhood.

Surprisingly, the concentration of immigrants in the current neighbourhood affects the earnings capacity of immigrants positively. This could be interpreted as an 'ethnic good' or 'ethnic network' effect.

 $^{^{21}}$ In this study, we ignore potential selectivity effects due to missing information on wages for non-employed, see Husted et al. (2001).

²² A similar conclusion is drawn by Österberg (2000), who concludes that Sweden - in contrast to the US - is characterised by a low cross-sectional income inequality and a high intergenerational income mobility.

Figure 6. Distribution of hourly wage rates in first job.



Province



	Dependent variable= log hourly wage rate, DKK					
	Mal	les	Fema	ales		
	2 nd generation immigrant	Ethnic Danes	2 nd generation immigrant	Ethnic Danes		
Parental variables:						
Years since migration, parents/100 ¹⁾	0.151 (0.267)	-	0.354 (0.283)	-		
Education, years, parents/100 ¹⁾	-0.600 (0.373)	-0.207* (0.092)	0.265 (0.332)	-0.367* (0.086)		
Log annual gross income, parents/100 ¹⁾	-0.438 (1.314)	-0.900* (0.399)	-0.223 (1.358)	-0.083* (0.345)		
Experience, years, parents/100 ¹⁾	0.057 (0.240)	-0.037 (0.054)	-0.164 (0.211)	-0.096 (0.050)		
Ethnic capital variables:				-		
Average years since migration/10	0.346 (0.307)	-	-0.541* (0.253)			
Average education/100 ²⁾	2.229 (2.331)	-	-4.751* (2.006)	-		
Average log annual gross income/100	-8.233 (25.821)	-	49.841* (22.374)	-		
Average years of experience/100	1.808 (2.200)	-	1.563 (1.882)	-		
Neighbourhood variables:						
Ethnic concentration in childhood municipality ³⁾	-1.742* (0.726)	0.364* (0.184)	-0.887 (0.635)	-0.041 (0.159)		
Exposure index, childhood municipality $^{\rm 4)}$	0.128 (0.191)	-	0.007 (0.012)	-		
Ethnic concentration in current municipality ³⁾	1.567* (0.657)	-0.202 (0.141)	0.991 (0.580)	0.011 (0.121)		
Exposure index, current municipality 4)	-0.013 (0.014)	-	0.003 (0.013)	-		
Other variables:	、 <i>,</i> ,					
Constant	4.295 (2.785)	3.564* (0.053)	-1.499 (2.418)	3.694* (0.048)		
Experience	0.035* (0.017)	0.047* (0.003)	0.026 (0.016)	0.016* (0.004)		
Experience squared/100	-0.028 (0.137)	-0.146* (0.027)	0.024 (0.146)	-0.047 (0.029)		
Education, years	0.043*	0.087*	0.053*	0.067*		
Married to an ethnic Dane ⁵⁾	0.148*	0.015	0.112	-0.007		
Married to an immigrant	-0.014	(0.019)	0.115	(0.012)		
	(0.176)		(0.136)			
Indicators for racion of critic	yes	yes	yes	yes		
Indicators for Degich and	yes	-	yes	-		
indicators for Danish regions	yes	yes	yes	yes		
K-squared	0.139	0.245	0.155	0.212		
No. of observations	1105	16554	1050	16628		

Table 5. OLS estimation of wage equation for 1997, individuals aged 20-35 years.

Notes: See Table 2.

For both males and females, the coefficients to the traditional human capital variables indicate that the human capital of young ethnic Danes is better remunerated than that of on 2^{nd} generation

immigrants.²³ The experience profiles are flatter for immigrants than for ethnic Danes (the experience coefficient of immigrant women is numerically larger than for Danish women, but is insignificant). The effect of one extra year of education is much lower for 2nd generation immigrants, especially males, than for ethnic Danes.²⁴ On average, the remuneration to an ethnic Dane is 9 % for a man and 7% for a woman while the same figures for 2nd generation immigrants are only 4 % and 5 %, respectively. This may be due to differences in the quality and type of education that 2nd generation immigrants and ethnic Danes acquire. But another obvious explanation is that it reflects discriminatory tendencies in the Danish labour market, see Nielsen et al. (2001). Finally, the marriage indicators show a positive effect of being married to an ethnic Dane, which may be interpreted as a positive network or language effect.

8. Conclusion

In this paper, we investigate the transition from school to work for 2^{nd} generation immigrants in Denmark. Analysing the school-to-work transition, we investigate four different aspects of this transition: the completion of a qualifying education, the waiting time before entry into the first ordinary job, the duration of the first employment spell and the hourly wage earned in 1997. The school-to-work transition of 2^{nd} generation immigrants is less successful than it is for young ethnic Danes. On average, they are less likely to obtain a qualifying education, the waiting time for the first job is longer, the first employment spell is shorter and their wages in 1997 are lower.

The focus is placed on the intergenerational transmission mechanisms, and the three main hypotheses that are tested concern the importance of parental capital, ethnic capital and neighbourhood effects for the school-to-work transition. In general, we are able to confirm that all three factors are correlated with successful labour market entry, although the strongest transmission mechanisms work through parental capital and childhood neighbourhoods. However, the influence on different aspects of the transition from school to work varies considerably. In general, the strongest effects of intergenerational transmissions happen in the educational attainment of the 2^{nd} generation immigrants, which again is very important for the employment experience of 2^{nd} generation immigrants, exactly as it is the case for young ethnic Danes. Also the waiting time for the first job after leaving the educational system is significantly affected by intergenerational effects, but the effects die out after the individuals have entered the labour market. This might be because demand side effects dominate supply side effects at this stage of the labour market career, and intergenerational transmissions are mainly thought to work through the individual decision making, which is on the supply side. In addition, there *is* a strong

²³ And the coefficients are unchanged if the intergenerational variables are left out.

²⁴ A similar result is reported by Österberg (2000) for Swedish males.

indirect intergenerational effect, which works through the strong effects of obtaining a qualifying education on employment duration and wages, and through the strong effect of education on the probability of obtaining a job in the first place.

Among the variables reflecting parental and ethnic capital, both years since migration and labour market experience are important. Hence, it does not only matter how long time the parents (or their generation) have been in the country, but also whether they have been attached to the labour market. This result leads to the conclusion that the 'assimilation process' of the descendants of 1st generation immigrants is not a process that 'automatically' takes place, given time. The 'assimilation process' is also dependent on the employment success of the parent generation, and if the parents are not successfully integrated into the labour market, this result may carry over to the children.

We find striking gender differences in the intergenerational transmission effects among 2nd generation immigrants. The overall picture from the analysis of the four aspects of the school-towork transition shows that the dominant intergenerational effect for women comes from the parents, whereas men are affected by ethnic capital and childhood neighbourhoods as well. Among the significant parental variables are years since migration, gross income and experience in the Danish labour market and therefore, we may conclude that labour market assimilation of the parents is extremely important for successful labour market entry of female 2nd generation immigrants. This may indicate that to some extent well-assimilated parents have adopted Danish norms with respect to female labour market participation. When it comes to the males, their labour market success is also affected by the concentration of immigrants in the neighbourhood where they grew up and furthermore, after getting a qualifying education, it is important whether they belong to an ethnic group who have been attached to the Danish labour market for many years.

Especially, in the analysis of the waiting time for the first job, the gender difference in the effect of parental capital is large. This may indicate that the parents 'allow' their daughters to take an education and then, potential 'parental gender discrimination' sets in when the daughter considers entering the labour market. This 'discrimination' may stem from arranging for the daughter to get married (with a man of similar nationality) and be a full-time housewife instead of entering the labour market. A related story may be told based on the coefficients to the marriage indicators. In almost all cases, we find a favourable effect from being married to an ethnic Dane, but for female 2nd generation immigrants, we find that being married to an immigrant reduces the probability of completing a qualifying education and the probability of entering the first job. Although, marriage is potentially endogenous, this result indicates an important problem for

young immigrant women, since only few young immigrant women marry an ethnic Dane.

As mentioned above, we find strong intergenerational transmissions in the attainment of a qualifying education. This is an important result because education turns out to be essential in all the studied phases of the school-to-work transition for all four groups in the study. Therefore, having parents with unfavourable characteristics, belonging to an unfavourable ethnic group or growing up in a neighbourhood with a high concentration of immigrants, is associated with a lower probability of achieving a qualifying education. As a consequence, the negative intergenerational influence potentially affects the whole labour market career not only directly, but also indirectly, through the attainment of education.

Since obtaining a qualifying education is shown to be extremely important for a successful integration into the labour market, it is unfortunate that the economic incentives for 2nd generation immigrants to acquire an education seem to be much lower than for young ethnic Danes. In the estimation of a simple wage equation, we find that the returns to both education and experience are considerably lower for 2nd generation immigrants compared to young ethnic Danes. In our study, we do not correct for self-selection into wage employment and neither do we control for potential differences in the quality or types of education that 2nd generation immigrants and ethnic Danes acquire, except for the effects controlled for by neighbourhood effects. Hence, it could be the case that the differences in the parameter estimates are due to stronger sample selection effects among 2nd generation immigrants relative to young ethnic Danes. However, the large differences in returns to educational investments are hard to explain without reference to theories of discrimination.

This leads us to the obvious policy conclusion that - to the extent that we are capturing causal relationships - schooling is important for young 2^{nd} generation immigrants, even if the returns to education in terms of hourly wages are lower for 2^{nd} generation immigrants than they are for young ethnic Danes. Hence, additional resources targeted at 2^{nd} generation immigrants in primary and secondary schools to induce them to undertake further education should be considered. Furthermore, all policy initiatives that potentially increase the returns to education for 2^{nd} generation immigrants should be considered.

References

Becker, G. S. and N. Tomes (1986), Human capital and the rise and fall of families, Journal of Labor Economics, vol. 4, pp. S1-39.

Borjas, G. J. (1992), Ethnic Capital and Intergenerational Mobility, Quarterly Journal of Economics, vol. 107(1), pp. 123-150.

Borjas, G. J. (1994), Immigrants skills and ethnic spillovers, Journal of Population Economics, 7 (2), pp. 99-118

Borjas, G. J. (1995), Ethnicity, Neighborhoods, and Human Capital Externalities, American Economic Review, vol. 85, pp. 365-390.

Borjas, G. J. (1999), Heaven's Door, Princeton UP, Princeton.

Card, D., J. DiNardo and E. Estes (1998), The more things change: Immigrants and the children of immigrants in the 1940s, the 1970s, and the 1990s, NBER working paper 6519, National Bureau of Economic Research, Massachusetts.

Coleman, J. S. (1988), Social Capital in the Creation of Human Capital, American Journal of Sociology, vol XCIV, pp. S95-120.

Edin, P.- A., P. Frederiksson, and O. Åslund (2000), Ethnic Enclaves and the Economic Succes of Immigrants - Evidence from a Natural Experiment, WP 2000:21, Dept. of Economics, University of Uppsala, Sweden.

Ekberg, J. (1997), Hur är arbetsmarknaden för den andra generationens invandrare?, Arbetsmarknad & Arbetsliv, vol. 3 (2), pp. 5-16.

Ginther, D., R. Haveman, and B. Wolfe (2000), Neighborhood Attributes as Determinants of Children's Outcomes: How Robust are the Relationships?, Journal of Human Resources 35 (4) pp. 603-642.

Hummelgård, H., B. K. Graversen, L. Husted and J. B. Nielsen (1998), Education and unemployment among young immigrants (in Danish), AKF rapport, AKF forlaget, Copenhagen

Husted, L., H. S. Nielsen, M. Rosholm and N. Smith (2001), Employment and Wage Assimilation of Male First Generation Immigrants in Denmark, forthcoming in International Journal of Manpower, vol. 22.

Jovanovich, B. (1979), Job Matching and the Theory of Turnover, Journal of Political Economy, vol. 87 (5), pp. 972-990.

Lancaster, T. (1990), The Econometric Analysis of Transition Data. Cambridge UP, Cambridge.

Mørkeberg, H. (2000), The education of immigrants, Statistics Denmark, Copenhagen.

Nielsen, H. J. (2000), The relationship between ethnic groups, Chapter 11 in Mogensen, G. V. and P. C. Matthiessen (eds.), Integration in Denmark (in Danish), Aarhus UP, Aarhus.

Nielsen, H. S., L. Husted, M. Rosholm, and N. Smith (2001), Qualifications, Discrimination or Assimilation: A New Framework for Analysing Immigrant Wage Gaps. Draft.

OECD (1999), Trends in International Migration, OECD, Paris.

Österberg, T. (2000), Economic Perspectives on Immigrants and Intergenerational Transmissions, Ekonomiska Studier 102, Göteborgs Universitet, Sweden.

Pedersen, S. (2000), The demography of Immigrants (in Danish), Chapter 1 in Mogensen, G. V. and P. C. Matthiessen (eds.), Integration in Denmark (in Danish), Aarhus UP, Aarhus.

Solon, G. (1999), Intergenerational mobility in the labour market, in Ashenfelter, O. and D. Card (eds.), Handbook of Labor Economics, Vol 3, Elsevier.

Appendix

Table A1. Mean sample values for 2^{nd} generation immigrants and ethnic Danes aged 18-35 years in 1997.

	Ma	les	Females		
	2 nd generation immigrant	Ethnic Danes	2 nd generation immigrant	Ethnic Danes	
Fraction who has completed a formal education	0.325 (0.468)	0.647 (0.478)	0.353 (0.478)	0.617 (0486)	
Duration of spell until first job, months	4.117 (9.079)	2.737 (7.915)	5.418 (10.930)	3.897 (9.789)	
Duration of first employment spell, months	14.166 (15.738)	19.927 (20.620)	14.411 (15.899)	19.183 (20.553)	
Log hourly wage rate, DKK 1997 prices	4.600 (0.542)	4.644 (0.494)	4.532 (0.462)	4.566 (0.451)	
Parental variables: Years since migration parents ¹⁾	14.348 (12.749)	-	14.542 (13.012)	-	
Education parents, years ¹⁾	6.845 (5.517)	10.403 (4.614)	7.864 (5.369)	10.380 (4.635)	
Log annual gross income, parents 1)	11.371 (2.915)	12.103 (1.998)	11.232 (3.222)	12.073 (2.085)	
Experience, years, parents ¹⁾	15.335 (9.055)	-	15.462 (9.076)	-	
Neighbourhood variables: Ethnic concentration in childhood municipality ³⁾	0.081 (0.046)	0.032 (0.031)	0.080 (0.046)	0.032 (0.031)	
Exposure index, childhood municipality $^{\rm 4)}$	2.744 (2.376)	-	2.701 (2.357)	-	
Ethnic concentration in current municipality ³⁾	0.107 (0.047)	0.066 (0.044)	0.108 (0.047)	0.067 (0.044)	
Exposure index, current municipality ⁴⁾	2.938 (2.338)	-	2.984 (2.323)	-	
<i>Ethnic capital variables:</i> Average years since migration ²⁾	11.831 (1.090)	-	11.799 (1.123)	-	
Average education ²⁾	6.568 (1.427)	-	6.607 (1.443)	-	
Average log annual gross income ²)	11.980 (0.159)	-	11.982 (0.158)	-	
Average years of experience ²⁾	10.132 (1.103)	-	10.162 (1.129)	-	
Other variables: Age	23.346 (4.888)	26.770 (5.098)	23.363 (4.945)	26.769 (5.111)	
Experience, years	3.215 (4.080)	7.309 (5.550)	2.686 (3.509)	5.650 (4.765)	
Education, years	10.569 (3.511)	11.548 (2.761)	10.779 (3.582)	11.826 (2.635)	
Married to an ethnic Dane ⁵⁾	0.095 (0.293)	0.194 (0.395)	0.109 (0.311)	0.273 (0.445)	
Married to an immigrant	0.007 (0.086)	-	0.026 (0.159)	-	
Indicator for missing years since migration for parents	0.380 (0.485)	-	0.379 (0.485)	-	
Indicator for other missing information on parents	0.054 (0.225)	0.019 (0.135)	0.058 (0.233)	0.021 (0.143)	
No. of observations	4579	66130	4345	63741	

Notes: See Table 2.

	Turkey	Pakistan	Other less developed countries	EU-12	Nordic	Other developed countries	Ethnic Danes
Age	20.0	20.4	21.7	27.4	26.9	23.4	26.8
	20.1	20.5	21.6	27.6	27.1	23.4	26.8
Education, years	9.7	10.5	10.7	11.9	10.9	10.7	11.7
	10.3	10.5	11.0	11.9	11.3	11.1	11.9
Hourly wage rate, DKK ¹⁾	125.6	127.3	142.9	167.3	172.8	147.9	156.3
	111.8	102.1	124.4	135.8	140.7	127.6	128.0
Annual gross income, thousand DKK	102.4	87.4	114.7	197.3	186.4	146.6	202.6
	99.2	77.6	104.0	154.4	151.7	129.4	154.1
Ethnic concentration ²⁾	10	13	12	9	9	11	7
	10	14	12	9	10	11	7
Exposure index ³⁾	2.8	5.0	0.1	0.7	0.5	0.5	1.0
	2.9	5.0	0.1	0.7	0.5	0.6	1.0
No. of observations	663	771	633	737	675	1019	66,130
	683	673	602	670	618	1013	63,741

Table A2. Mean sample values for 2^{nd} generation immigrants and ethnic Danes aged 18-35 years in 1997. (Values for males above, for females below).

1) The calculation of wage rates is restricted to individuals who are employed and not enrolled in the educational system.

2) Ethnic concentration is defined as the percentage of 1st and 2nd generation immigrants in the municipality concerned.
3) Exposure index is defined as the fraction of immigrants from own ethnic origin in the municipality concerned divided by the fraction in the total population from own ethnic origin.

IZA Discussion Papers

No	Author(s)	Titel	Area	Date
211	A. van Soest M. Das X. Gong	A Structural Labour Supply Model with Nonparametric Preferences	5	11/00
212	X. Gong A. van Soest P. Zhang	Sexual Bias and Household Consumption: A Semiparametric Analysis of Engel Curves in Rural China	5	11/00
213	X. Gong A. van Soest E. Villagomez	Mobility in the Urban Labor Market: A Panel Data Analysis for Mexico	1	11/00
214	X. Gong A. van Soest	Family Structure and Female Labour Supply in Mexico City	5	11/00
215	J. Ermisch M. Francesconi	The Effect of Parents' Employment on Children's Educational Attainment	5	11/00
216	F. Büchel	The Effects of Overeducation on Productivity in Germany — The Firms' Viewpoint	5	11/00
217	J. Hansen R. Wahlberg	Occupational Gender Composition and Wages in Sweden	5	11/00
218	C. Dustmann A. van Soest	Parametric and Semiparametric Estimation in Models with Misclassified Categorical Dependent Variables	1	11/00
219	F. Kramarz T. Philippon	The Impact of Differential Payroll Tax Subsidies on Minimum Wage Employment	5	11/00
220	W. A. Cornelius E. A. Marcelli	The Changing Profile of Mexican Migrants to the United States: New Evidence from California and Mexico	1	12/00
221	C. Grund	Wages as Risk Compensation in Germany	5	12/00
222	W.P.M. Vijverberg	Betit: A Family That Nests Probit and Logit	7	12/00
223	M. Rosholm M. Svarer	Wages, Training, and Job Turnover in a Search- Matching Model	1	12/00
224	J. Schwarze	Using Panel Data on Income Satisfaction to Estimate the Equivalence Scale Elasticity	3	12/00

225	L. Modesto J. P. Thomas	An Analysis of Labour Adjustment Costs in Unionized Economies	1	12/00
226	P. A. Puhani	On the Identification of Relative Wage Rigidity Dynamics: A Proposal for a Methodology on Cross-Section Data and Empirical Evidence for Poland in Transition	4/5	12/00
227	L. Locher	Immigration from the Eastern Block and the former Soviet Union to Israel: Who is coming when?	1	12/00
228	G. Brunello S. Comi C. Lucifora	The College Wage Gap in 10 European Countries: Evidence from Two Cohorts	5	12/00
229	R. Coimbra T. Lloyd-Braga L. Modesto	Unions, Increasing Returns and Endogenous Fluctuations	1	12/00
230	L. Modesto	Should I Stay or Should I Go? Educational Choices and Earnings: An Empirical Study for Portugal	5	12/00
231	G. Saint-Paul	The Economics of Human Cloning	5	12/00
232	E. Bardasi M. Francesconi	The Effect of Non-Standard Employment on Mental Health in Britain	5	12/00
233	C. Dustmann C. M. Schmidt	The Wage Performance of Immigrant Women: Full-Time Jobs, Part-Time Jobs, and the Role of Selection	1	12/00
234	R. Rotte M. Steininger	Sozioökonomische Determinanten extremistischer Wahlerfolge in Deutschland: Das Beispiel der Eu- ropawahlen 1994 und 1999	3	12/00
235	W. Schnedler	Who gets the Reward? An Empirical Exploration of Bonus Pay and Task Characteristics	5	12/00
236	R. Hujer M. Caliendo	Evaluation of Active Labour Market Policy: Methodological Concepts and Empirical Estimates	6	12/00
237	S. Klasen I. Woolard	Surviving Unemployment without State Support: Unemployment and Household Formation in South Africa	3	12/00
238	R. Euwals A. Börsch-Supan A. Eymann	The Saving Behaviour of Two Person House- holds: Evidence from Dutch Panel Data	5	12/00

239	F. Andersson K. A. Konrad	Human Capital Investment and Globalization in Extortionary States	5	01/01
240	W. Koeniger	Labor and Financial Market Interactions: The Case of Labor Income Risk and Car Insurance in the UK 1969-95	5	01/01
241	W. Koeniger	Trade, Labor Market Rigidities, and Government- Financed Technological Change	2	01/01
242	G. Faggio J. Konings	Job Creation, Job Destruction and Employment Growth in Transition Countries in the 90's	4	01/01
243	E. Brainerd	Economic Reform and Mortality in the Former Soviet Union: A Study of the Suicide Epidemic in the 1990s	4	01/01
244	S. M. Fuess, Jr. M. Millea	Pay and Productivity in a Corporatist Economy: Evidence from Austria	5	01/01
245	F. Andersson K. A. Konrad	Globalization and Human Capital Formation	5	01/01
246	E. Plug W. Vijverberg	Schooling, Family Background, and Adoption: Does Family Income Matter?	5	01/01
247	E. Plug W. Vijverberg	Schooling, Family Background, and Adoption: Is it Nature or is it Nurture?	5	01/01
248	P. M. Picard E. Toulemonde	The Impact of Labor Markets on Emergence and Persistence of Regional Asymmetries	2	01/01
249	B. M. S. van Praag P. Cardoso	"Should I Pay for You or for Myself?" The Optimal Level and Composition of Retirement Benefit Systems	3	01/01
250	T. J. Hatton J. G. Williamson	Demographic and Economic Pressure on Emigration out of Africa	1	01/01
251	R. Yemtsov	Labor Markets, Inequality and Poverty in Georgia	4	01/01
252	R. Yemtsov	Inequality and Income Distribution in Georgia	4	01/01
253	R. Yemtsov	Living Standards and Economic Vulnerability in Turkey between 1987 and 1994	4	01/01
254	H. Gersbach A. Schniewind	Learning of General Equilibrium Effects and the Unemployment Trap	3	02/01
255	H. Gersbach A. Schniewind	Product Market Reforms and Unemployment in Europe	3	02/01

256	T. Boeri H. Brücker	Eastern Enlargement and EU-Labour Markets: Perceptions, Challenges and Opportunities	2	02/01
257	T. Boeri	Transition with Labour Supply	4	02/01
258	M. Rosholm K. Scott L. Husted	The Times They Are A-Changin': Organizational Change and Immigrant Employment Opportunities in Scandinavia	1	02/01
259	A. Ferrer-i-Carbonell B. M.S. van Praag	Poverty in the Russian Federation	4	02/01
260	P. Cahuc F. Postel-Vinay	Temporary Jobs, Employment Protection and Labor Market Performance	1/3	02/01
261	M. Lindahl	Home versus School Learning: A New Approach to Estimating the Effect of Class Size on Achievement	5	02/01
262	M. Lindahl	Summer Learning and the Effect of Schooling: Evidence from Sweden	5	02/01
263	N. Datta Gupta N. Smith	Children and Career Interruptions: The Family Gap in Denmark	5	02/01
264	C. Dustmann	Return Migration, Wage Differentials, and the Optimal Migration Duration	1	02/01
265	M. Rosholm M. Svarer	Structurally Dependent Competing Risks	1	02/01
266	C. Dustmann O. Kirchkamp	The Optimal Migration Duration and Activity Choice after Re-migration	1	02/01
267	A. Newell	The Distribution of Wages in Transition Countries	4	03/01
268	A. Newell B. Reilly	The Gender Pay Gap in the Transition from Communism: Some Empirical Evidence	4	03/01
269	H. Buddelmeyer	Re-employment Dynamics of Disabled Workers	3	03/01
270	B. Augurzky C. M. Schmidt	The Evaluation of Community-Based Interventions: A Monte Carlo Study	6	03/01
271	B. Augurzky C. M. Schmidt	The Propensity Score: A Means to An End	6	03/01

272	C. Belzil J. Hansen	Heterogeneous Returns to Human Capital and Dynamic Self-Selection	5	03/01
273	G. Saint-Paul	Distribution and Growth in an Economy with Limited Needs	5	03/01
274	P. J. Pedersen N. Smith	Unemployment Traps: Do Financial Dis- incentives Matter?	3	03/01
275	G. S. Epstein T. Lecker	Multi-Generation Model of Immigrant Earnings: Theory and Application	1	03/01
276	B. Amable D. Gatti	The Impact of Product Market Competition on Employment and Wages	5	03/01
277	R. Winter-Ebmer	Evaluating an Innovative Redundancy-Retraining Project: The Austrian Steel Foundation	6	03/01
278	T. M. Andersen	Welfare Policies, Labour Taxation and Inter- national Integration	2	04/01
279	T. M. Andersen	Product Market Integration, Wage Dispersion and Unemployment	2	04/01
280	P. Apps R. Rees	Household Saving and Full Consumption over the Life Cycle	7	04/01
281	G. Saint-Paul	Information Technology and the Knowledge Elites	5	04/01
282	J. Albrecht A. Björklund S. Vroman	Is There a Glass Ceiling in Sweden?	5	04/01
283	M. Hagedorn A. Kaul V. Reinthaler	Welfare Analysis in a Schumpeterian Growth Model with Capital	7	04/01
284	H. Rapoport A. Weiss	The Optimal Size for a Minority	1	04/01
285	J. Jerger C. Pohnke A. Spermann	Gut betreut in den Arbeitsmarkt? Eine mikroökonometrische Evaluation der Mannheimer Arbeitsvermittlungsagentur	5	04/01
286	M. Fertig C. M. Schmidt	First- and Second-Generation Migrants in Germany –What Do We Know and What Do People Think	1	04/01

287	P. Guggenberger A. Kaul M. Kolmar	Efficiency Properties of Labor Taxation in a Spatial Model of Restricted Labor Mobility	3	04/01
288	D. A. Cobb-Clark	Getting Ahead: The Determinants of and Payoffs to Internal Promotion for Young U.S. Men and Women	5	04/01
289	L. Cameron D. A. Cobb-Clark	Old-Age Support in Developing Countries: Labor Supply, Intergenerational Transfers and Living Arrangements	3	04/01
290	D. A. Cobb-Clark M. D. Connolly C. Worswick	The Job Search and Education Investments of Immigrant Families	1	04/01
291	R. T. Riphahn	Cohort Effects in the Educational Attainment of Second Generation Immigrants in Germany: An Analysis of Census Data	1	05/01
292	E. Wasmer	Between-group Competition in the Labor Market and the Rising Returns to Skill: US and France 1964-2000	5	05/01
293	D. Cobb-Clark T. F. Crossley	Gender, Comparative Advantage and Labor Market Activity in Immigrant Families	1	05/01
294	Š. Jurajda	Estimating the Effect of Unemployment Insurance Compensation on the Labor Market Histories of Displaced Workers	3	05/01
295	F. Duffy P. P. Walsh	Individual Pay and Outside Options: Evidence from the Polish Labour Force Survey	4	05/01
296	H. S. Nielsen M. Rosholm N. Smith L. Husted	Intergenerational Transmissions and the School- to-Work transition of 2 nd Generation Immigrants	1	05/01